Author’s response to reviews

Title: Teaching evidence based medicine literature searching skills to medical students during the clinical years - a protocol for a randomised controlled trial

Authors:

Dragan Ilic (dragan.ilic@med.monash.edu.au)
Katrina Tepper (katrina.tepper@monash.edu)
Marie Misso (marie.misso@monash.edu)

Version: 4 Date: 8 July 2011

Author’s response to reviews: see over
8th July 2011

Dear Editor,

MS: 7668651351048784
Teaching evidence based medicine literature searching skills to medical students during the clinical years – a protocol for a randomised controlled trial

We thank you for accepting our above titled protocol for publication in BMC Medical Education. We have read through the minor essential revision provided by the reviewer and have amended our manuscript accordingly.

We have also read through the manuscript and ensured that it conforms to all points of the formatting checklist as requested.

We thank you again for the opportunity to publish in BMC Medical Education and look forward to your future correspondence.

Kind regards,

Dr Dragan Illic
Senior Lecturer
Department of Epidemiology & Preventive Medicine
School of Public Health & Preventive Medicine
Monash University
The Alfred Centre
Level 6, 99 Commercial Road
Melbourne VIC 3004
Reviewer’s Report:
Minor Essential
Please indicate, on Figure 1, the time scales: How soon (long) after the EBM-LSS workshop will the Fresno/EBPQ assessments be made (in both conditions—before and during/?after the 3rd year)
We have insert the following sentence on page 7 of the manuscript under ‘Outcome measures’ as suggested – “These outcomes will be measured across both intervention and control groups 1-week post-implementation of the intervention”.

We have also indicated this in Figure 1 as suggested.